Author's response to reviews

Title: Manual cervical distraction for neck pain: results of a pilot randomized controlled trial

Authors:

Maruti R Gudavalli (Gudavalli_r@palmer.edu)
Stacie A Salsbury (stacie.salsbury@palmer.edu)
Robert D Vining (robert.vining@palmer.edu)
Cynthia R Long (LONG_C@palmer.edu)
Lance Corber (CORBER_L@palmer.edu)
Avinash G Patwardhan (apatwar@lumc.edu)
Christine M Goertz (christine.goertz@palmer.edu)

Version: 2 Date: 13 November 2014

Author’s response to reviews:

Format changes as requested by the Editorial office.
1. Included the date of trial registration
2. Removed the Trial Status Section
3. Moved the Figure legend and tables below the reference list
4. Consent section added for use of images
5. Faces in the images are masked